Development of a core outcome set for disease modification trials in mild to moderate dementia: a systematic review, patient and public consultation and consensus recommendations

Lucy Webster,¹ Derek Groskreutz,²
Anna Grinbergs-Saull,³ Rob Howard,¹
John T O'Brien,⁴ Gail Mountain,⁵ Sube Banerjee,⁶
Bob Woods,⁷ Robert Perneczky,⁸ Louise Lafortune,⁹
Charlotte Roberts,¹⁰ Jenny McCleery,¹¹
James Pickett,³ Frances Bunn,¹² David Challis,¹³
Georgina Charlesworth,¹⁴ Katie Featherstone,¹⁵
Chris Fox,¹⁶ Claire Goodman,¹² Roy Jones,¹⁷
Sallie Lamb,¹⁸ Esme Moniz-Cook,¹⁹ Justine Schneider,²⁰
Sasha Shepperd,²¹ Claire Surr,²² Jo Thompson-Coon,²³
Clive Ballard,²⁴ Carol Brayne,⁹ Orlaith Burke,²¹
Alistair Burns,²⁵ Linda Clare,^{23,26,27} Peter Garrard,²⁸
Patrick Kehoe,²⁹ Peter Passmore,³⁰ Clive Holmes,³¹
lan Maidment,³² Fliss Murtagh,³³ Louise Robinson³⁴
and Gill Livingston^{1,35,36}*

¹Division of Psychiatry, University College London, London, UK

²Division of Psychology and Language Sciences, University College London, London, UK

³Alzheimer's Society, London, UK

⁴Department of Psychiatry, University of Cambridge, Cambridge, UK

⁵School of Health and Related Research, University of Sheffield, Sheffield, UK

⁶Brighton and Sussex Medical School, University of Sussex, Brighton, UK

⁷Dementia Services Development Centre Wales, Bangor University, Bangor, UK

⁸Faculty of Medicine, School of Public Health, Imperial College London, London, UK

⁹Cambridge Institute of Public Health, University of Cambridge, Cambridge, UK ¹⁰International Consortium for Health Outcomes Measurement, London, UK

¹¹Oxford Health NHS Foundation Trust, Banbury, UK

- ¹²Centre for Research in Primary and Community Care, University of Hertfordshire, Hatfield, UK
- ¹³Personal Social Services Research Unit, University of Manchester, Manchester, UK
- ¹⁴Research Department of Clinical, Educational, and Health Psychology, University College London, London, UK
- ¹⁵School of Healthcare Sciences, Cardiff University, Cardiff, UK
- ¹⁶Norwich Medical School, University of East Anglia, Norwich, UK
- ¹⁷Research Institute for the Care of Older People, University of Bath, Bath, UK
- ¹⁸Oxford Clinical Trials Research Unit, University of Oxford, Oxford, UK
- ¹⁹Faculty of Health and Social Care, University of Hull, Hull, UK
- ²⁰Institute of Mental Health, University of Nottingham, Nottingham, UK
- ²¹Nuffield Department of Population Health, University of Oxford, Oxford, UK
- ²²School of Health & Community Studies, Leeds Beckett University, Leeds, UK
- ²³Collaboration for Leadership in Applied Health Research and Care South West Peninsula, University of Exeter, Exeter, UK
- ²⁴Wolfson Centre for Age-Related Diseases, King's College London, London, UK
- ²⁵Institute of Brain, Behaviour and Mental Health, University of Manchester, Manchester, UK
- ²⁶School of Psychology, University of Exeter, Exeter, UK
- ²⁷Centre for Research in Ageing and Cognitive Health, University of Exeter Medical School, Exeter, UK
- ²⁸Neuroscience Research Centre, St George's, University of London, UK
- ²⁹School of Clinical Sciences, University of Bristol, Bristol, UK
- ³⁰Centre for Public Health, Queen's University Belfast, Belfast, UK
- ³¹School of Medicine, University of Southampton, Southampton, UK
- ³²Aston Research Centre for Healthy Ageing, Aston University, Birmingham, UK
- ³³Cicely Saunders Institute, King's College London, London, UK
- ³⁴Institute of Health and Society, Newcastle University, Newcastle upon Tyne, UK
- ³⁵Camden and Islington NHS Foundation Trust, London, UK
- ³⁶North Thames Collaboration for Leadership in Applied Health Research and Care, London, UK

Declared competing interests of authors: Clive Ballard reports grants and personal fees from Lundbeck Ltd and ACADIA Pharmaceuticals Inc., and personal fees from Roche, Orion Pharma, GlaxoSmithKline, Otsuka Pharmaceutical, Heptares Therapeutics Ltd and Eli Lilly and Company outside the submitted work. Sube Banerjee reports grants and personal fees from AbbVie, personal fees and non-financial support from Eli Lilly and Company, and personal fees from Eleusis Pharmaceuticals Ltd, Daval International, Boehringer Ingelheim, Axovant Sciences, Lundbeck Ltd and Nutricia outside the submitted work. Sube Banerjee also reports being a member of the Health Technology Assessment (HTA) Mental, Psychological and Occupational Health Panel, and has been involved as the principal investigator in a series of National Institute for Health Research grants that has developed the Dementia Quality of Life measure (DEMQOL) system for the measurement of health-related quality of life in dementia, which is one of the candidate measures in this project. Alistair Burns reports being the editor for the International Journal of Geriatric Psychiatry and being the National Clinical Director for Dementia, NHS England, during the conduct of the study. Peter Garrard reports personal fees from Merck Sharp and Dohme Ltd outside the submitted work. Esme Moniz-Cook reports non-financial support from her contribution to the recent Joint Programme – Neurodegenerative Disease Research Outcome Measures, outside the submitted work. John T O'Brien reports personal fees from GE Healthcare, TauRx Pharmaceuticals, Cytox Ltd and Accera Inc, and grants and personal fees from Avid Radiopharmaceuticals

^{*}Corresponding author G.livingston@ucl.ac.uk

Inc./Eli Lilly and Company outside the submitted work. James Pickett reports being a full-time employee of the Alzheimer's Society. Rob Howard is a member of the HTA Commissioning Board. Sasha Shepperd is a member of the Health Services and Delivery Research Prioritisation Commissioning Panel.

Published May 2017 DOI: 10.3310/hta21260

Plain English summary

Disease modification trials in mind to moderate dementia

Health Technology Assessment 2017; Vol. 21: No. 26

DOI: 10.3310/hta21260

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

n the UK, around 850,000 people have dementia. If a treatment can change the underlying pathology of dementia this is called disease modification, although no trials have yet found effective disease-modifying treatments. Trials have used differing outcome measures to evaluate if a treatment works, making it difficult to compare and contrast results. To address this issue we aimed, in collaboration with the UK dementia research community and the Alzheimer's Society's Research Network, to develop a core set of outcome measures for use in future disease-modifying trials for mild to moderate dementia.

We looked at the outcomes used across completed and ongoing disease modification trials and found measures in six test areas: cognition, biological, behaviour, quality of life, activities of daily living and global. We used these findings to conduct a small consultation with people living with dementia and family carers. We presented all results at our consensus conference and discussed them to reach our conclusions.

We recommend that the core set of outcome measures should include a cognitive measure, namely the Mini Mental State Examination or the Alzheimer's Disease Assessment Scale – Cognitive subscale, and an optional magnetic resonance imaging scan looking at brain structure as a biological measure. We have specified measures for the other areas that are important but not core. The recommendations may change as new measures are developed, and, as most of the trials included participants with Alzheimer's disease only, recommendations need to be developed for different dementias. They apply only to mild to moderate stages of dementia.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.058

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the ISI Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 15/62/02. The contractual start date was in December 2015. The draft report began editorial review in June 2016 and was accepted for publication in October 2016. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2017. This work was produced by Webster et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Health Technology Assessment Editor-in-Chief

Professor Hywel Williams Director, HTA Programme, UK and Foundation Professor and Co-Director of the Centre of Evidence-Based Dermatology, University of Nottingham, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Health Sciences Research, Health and Wellbeing Research Group, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk